%42 % % 3
2023 4% 8 A

T
URANIUM MINING AND METALLURGY

% Vol. 42 No. 3
Aug. 2023

REYELEHET RHIZMR

EHF REE.H F

SERRA LS R EE

B Lok At st L & WF5E e - b st 101149)

W LW B AR B XE RS 2 R A T B AP B R A B X B B X A R
I3 HT T MERALRL L IFAE R N T R U B SR R AT TR R AT Y . A SRR v vk PR A B R
M 4 T R R R R 3 AL R e RN T AR BE A A TR . TR AR — 0. 15 mm (G
e 75%0) BRER A& 20 %0 MnO, & 200 B B EE 90 'C R HEE 3 h &4 AR B R b R REE B
300 ~40 0 H w2 7200 . WETEAE RONIZ BT B U A0 T A AR T A 0 i DR G 8 B 10 3R AOR BT

T
KW U AR A AL

FESES: TFI1L.3 XEFEES: A XEHS: 1000-8063(2023)03-0024-05

DOI: 10. 13426/j. cnki. yky. 2022. 12. 01

FHEA L R A AL o SBUBRCE BE - 7616 4
AT R A SR AT 2 )z B . b S R
il 12k o (EL B A 7 ARG L XE AR BE A R S 2 5
BV SETE AR MR A B T A
B 2 A5 WO R IR 25 Mo A28 Mo (VD) A i
B Mo CVD) ¥ fif M THT S5 B % 5H B 5 A9 1]
WA o SRS R S B A 87 90 L 0K 06 MR 4 D
14 5 T T R R A T L U SRR B SR A R R
FE A3 AR iR

F AT [ A M 5 22 46 Hp 7 i DR AR B 1
TS A 4 25 A AT A [ A0 00 36 T A0 2 L4
TR T 8D . E AR Bk R
REDONEAE= W ok SNEI RS i ob/=3 0D I3k E R = W S = 1)
B HEFE N R R R R A
Rebe — R FOARNY PSR T AR RGBT R
T2 BTN M A B BH B A 0 2 R A

TE 53 A i 249 BRI H 20RO B D R A SR o
58 A 35 R BRI ST o [P o R R I T ) A
B AE N A Tl A iR A2 %

1 T AESHE

A I IR AR B BT IR (T A S IR G
(R D ZH FET P2 H B W i A il H ik
55 LUBCR B W AA e E LGB S S v s 4. 0
AR AR 0. 21 %, e T2 8 W24 WF 58 45 1
W 2, FTLLE W Z0 AP YRR L
SR AR TE TR A oh, 3 80R A
B35 TR ME L AR X o i AT AR R . 0
Y AR 2 BT 2R B 20 A b B e B AN o
I 24,66 %0, AR AR RS AT O R BR AL 5 P
Hrp e b G SR 23, 21%, Kk,
B WA 3 AR R R BB T A T A

R1 THEELZERS

Table 1 Main chemical components of the mineral sample %
% Mo Fe Al SiO; As F Ca Mg S
o A8 0.21 3.52 6.91 71.0 0.01 0.13 <0.03 0.21 4,24
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Table 2 Process mineralogy results of the ore

FER TS w(Mo) /% A E) %
A PR AR 0. 030 14. 21
S 0. 022 10. 45
W 0. 049 23.21
JiE Hi 4R 0.110 52.13

/N 0.211 100. 00
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Table 3 Conditions and results of

conventional acid leaching method

g MnO ‘ = J4 it Mo 2
/% M/ C W/ %
—0.15  10% iR 2 il 33.79
—0.15  10% iR 2 e 40. 48
—0.15  10% iR 2 % 43.53
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Fig. 1 Effect of sulfuric acid dosage

/%

ol

on molybdenum leaching rate
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Fig. 2 Effect of MnO, dosage on molybdenum leaching rate
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Fig. 3 Effect of ore granularity on molybdenum
leaching rate
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Fig. 4 Effect of temperature on molybdenum

leaching rate
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Fig. 5 Effect of leaching time on molybdenum

leaching rate
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Study on Leaching Technology of a Refractory Molybdenum Ore
LIANG Gengyu, KANG Shaohui, MENG Jin, WANG Hongming, YANG Jianfei, WU Cuilian
(Beijing Research Institute of Chemical Engineering and Metallurgy, CNNC, Beijing 101149, China)

Abstract; A molybdenum ore has the characteristics of low grade, difficult separation, various molyb-
denum minerals and some of them are located in inclusions. In view of the refractory characteristics of
the molybdenum ore, the leaching mechanism was analyzed. At high temperature, the leaching test
was carried out with high concentration sulfuric acid. The results show that high concentration sulfu-
ric acid crash pyrite inclusions; increasing the amount of sulfuric acid, increasing the temperature, in-
creasing oxidant dosage and reducing the ore size are beneficial to the Mo leaching rate. Under the
conditions of particle size —0. 15 mm(accounting for 75%), sulfuric acid dosage 20% , MnQ, dosage
2%, leaching temperature 90 C, leaching time 3 h, the leaching rate of molybdenum increases from
30% ~40% of conventional stirring leaching to 72%. The research results put forward some counter-
measures for the exploitation and utilization of molybdenum ore resources, and made a breakthrough
in molybdenum leaching effect.

Key words: inclusion; molybdenum ore; enhanced leaching; oxidant





